Author's response to reviews

Title: The feminization of the medical work force, implications for Scottish primary care: a survey of Scottish general practitioners.

Authors:

Brian H Mckinstry (brian.mckinstry@ed.ac.uk)
Iain Colthart (iain.colthart@nes.scot.nhs.uk)
Katy Eliott (katy.elliott@nes.scot.nhs.uk)
Colin M Hunter (colin.m.hunter@yahoo.co.uk)

Version: 4  Date: 23 April 2006

Author's response to reviews:

My apologies, unfortunately I had not altered the conclusion in the main part of the paper in line with the abstract. This has now been done and meets the concerns of reviewer 2.

I have provided an explanation for the term 'job-share' under table 1. This type of post no longer exists but a few doctors are still in contracts set up previously. Basically it refers to two doctors who chose to share the hours and responsibilities of a full-time partnership post in a flexible way. (As part-time posts and retainer posts, have become more readily available in recent years this type of post has become redundant)